

Noyce, AJ; Kia, DA; Hemani, G; Nicolas, A; Price, TR; De Pablo-Fernandez, E; Haycock, PC; Lewis, PA; Foltynie, T; Davey Smith, G; International Parkinson Disease Genomics Consortium, ; Schrag, A; Lees, AJ; Hardy, J; Singleton, A; Nalls, MA; Pearce, N; Lawlor, DA; Wood, NW (2017) Estimating the causal influence of body mass index on risk of Parkinson disease: A Mendelian randomisation study. PLoS medicine, 14 (6). e1002314. ISSN 1549-1277 DOI: https://doi.org/10.1371/journal.pmed.

Downloaded from: http://researchonline.lshtm.ac.uk/3962407/

DOI: 10.1371/journal.pmed.1002314

Usage Guidelines

 $Please \ refer \ to \ usage \ guidelines \ at \ http://researchonline.lshtm.ac.uk/policies.html \ or \ alternatively \ contact \ researchonline@lshtm.ac.uk.$ 

Available under license: http://creativecommons.org/licenses/by/2.5/



# 

**Citation:** Noyce AJ, Kia DA, Hemani G, Nicolas A, Price TR, De Pablo-Fernandez E, et al. (2017) Estimating the causal influence of body mass index on risk of Parkinson disease: A Mendelian randomisation study. PLoS Med 14(6): e1002314. https://doi.org/10.1371/journal.pmed.1002314

Academic Editor: Carol Brayne, University of Cambridge, UNITED KINGDOM

Received: November 12, 2016

Accepted: May 4, 2017

Published: June 13, 2017

**Copyright:** This is an open access article, free of all copyright, and may be freely reproduced, distributed, transmitted, modified, built upon, or otherwise used by anyone for any lawful purpose. The work is made available under the <u>Creative</u> Commons CC0 public domain dedication.

Data Availability Statement: All relevant data are within the paper and its Supporting Information files. GIANT consortium data are available at http:// portals.broadinstitute.org/collaboration/giant/index. php/GIANT\_consortium. IPDGC consortium data are available at http://www.pdgene.org/ and involves an application to be completed.

**Funding:** The authors received no specific funding for this work. Other individual financial disclosures: AJN is funded by Parkinson's UK (ref F1201). DAK is supported by an MB PhD Award from the

**RESEARCH ARTICLE** 

# Estimating the causal influence of body mass index on risk of Parkinson disease: A Mendelian randomisation study

Alastair J. Noyce<sup>1,2</sup>, Demis A. Kia<sup>1</sup>, Gibran Hemani<sup>3,4</sup>, Aude Nicolas<sup>5</sup>, T. Ryan Price<sup>5</sup>, Eduardo De Pablo-Fernandez<sup>1</sup>, Philip C. Haycock<sup>3,4</sup>, Patrick A. Lewis<sup>1,6</sup>, Thomas Foltynie<sup>7</sup>, George Davey Smith<sup>3,4</sup>, International Parkinson Disease Genomics Consortium<sup>1</sup>, Anette Schrag<sup>8</sup>, Andrew J. Lees<sup>1</sup>, John Hardy<sup>1</sup>, Andrew Singleton<sup>5</sup>, Mike A. Nalls<sup>5,9</sup>, Neil Pearce<sup>10</sup>, Debbie A. Lawlor<sup>3,4</sup>, Nicholas W. Wood<sup>1</sup>\*

1 Department of Molecular Neuroscience, UCL Institute of Neurology, University College London, London, United Kingdom, 2 Centre for Neuroscience and Trauma, Barts and The London School of Medicine and Dentistry, Queen Mary University of London, London, United Kingdom, 3 MRC Integrative Epidemiology Unit, University of Bristol, Bristol, United Kingdom, 4 School of Social and Community Medicine, University of Bristol, Bristol, United Kingdom, 5 Laboratory for Neurogenetics, National Institute on Aging National Institutes of Health, Bethesda, Maryland, United States of America, 6 School of Pharmacy, University of Reading, Reading, University College London, London, United Kingdom, 8 Department of Clinical Neurosciences, UCL Institute of Neurology, University College London, London, United Kingdom, 9 Data Tecnica International, Glen Echo, Maryland, United States of America, 10 Department of Medical Statistics, London School of Hygiene & Tropical Medicine, London, United Kingdom

• These authors contributed equally to this work.

 $\P$  Membership of the International Parkinson Disease Genomics Consortium is provided in the Acknowledgments.

\* n.wood@ucl.ac.uk

## Abstract

## Background

Both positive and negative associations between higher body mass index (BMI) and Parkinson disease (PD) have been reported in observational studies, but it has been difficult to establish causality because of the possibility of residual confounding or reverse causation. To our knowledge, Mendelian randomisation (MR)—the use of genetic instrumental variables (IVs) to explore causal effects—has not previously been used to test the effect of BMI on PD.

## Methods and findings

Two-sample MR was undertaken using genome-wide association (GWA) study data. The associations between the genetic instruments and BMI were obtained from the GIANT consortium and consisted of the per-allele difference in mean BMI for 77 independent variants that reached genome-wide significance. The per-allele difference in log-odds of PD for each of these variants was estimated from a recent meta-analysis, which included 13,708 cases of PD and 95,282 controls. The inverse-variance weighted method was used to estimate a pooled odds ratio (OR) for the effect of a 5-kg/m<sup>2</sup> higher BMI on PD. Evidence of directional pleiotropy averaged across all variants was sought using MR–Egger regression. Frailty



International Journal of Experimental Pathology. GH, PCH, GDS and DAL work in a unit that receives funding from the University of Bristol and the UK Medical Research Council (MC\_UU\_1201/1 and MC UU 1201/5). DAL is a National Institute of Health Research Senior Investigator (NF-SI-0166-10196). PCH is supported by CRUK Population Research Postdoctoral Fellowship C52724/ A20138. PAL receives funding from the MRC (MR/ N026004/1 and MR/L010933/1). MAN, ASi, AN and TRP participation in this study was supported in part by the Intramural Research Program of the National Institute on Aging, NIH. University College London Hospitals and University College London receive support from the Department of Health's National Institute for Health Research (NIHR) Biomedical Research Centres (BRC). NWW is an NIHR senior investigator and receives support from the JPND-MRC Comprehensive Unbiased Risk factor Assessment for Genetics and Environment in Parkinson's disease (COURAGE). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: AJN reports: grants from Parkinson's UK, Élan/Prothena Pharmaceuticals, GE Healthcare; shares in LifeLab Ltd; advisory board membership for mvHealthPal: honoraria from Office Octopus, Henry Stewart Talks, and Britannia Pharmaceuticals Ltd, all outside of the submitted work. PAL reports grants from MRC, BBSRC, National Institutes of Health, Michael J. Fox Foundation, personal fees from Astex Pharmaceuticals, outside the submitted work. TF reports grants from Michael J Fox Foundation, grants from Brain Research Trust, grants from European Union FP7, grants from John Black Charitable Foundation, personal fees from Medtronic, personal fees from BIAL, personal fees from Profile Pharma, personal fees from Brittania, all outside the submitted work. ASc reports grants from GE Healthcare, outside the submitted work. AJL reports personal fees from Britannia Pharmaceuticals Ltd, BIAL Portela, Profile Pharma Ltd, Teva, Lundbeck, Roche, UCB, NeuroDerm, Nordicinfu Care. Decision Resources. all outside of the submitted work. GDS is a member of the Editorial Board of PLOS Medicine. MAN is a paid consultant/contractor to the NIH. DAL reports grants from UK Medical Research Council, during the conduct of the study; grants from Medical Research council, grants from Wellcome Trust, grants from European Research Council, grants from Roche Diagnostics, grants from Medtronic, grants from Ferring Pharmaceuticals, outside the submitted work. DAK, GH, ASi, AN, EDP-F, JH, NP, TRP, NWW report no relevant disclosures.

simulations were used to assess whether causal associations were affected by mortality selection.

A combined genetic IV expected to confer a lifetime exposure of 5-kg/m<sup>2</sup> higher BMI was associated with a lower risk of PD (OR 0.82, 95% CI 0.69–0.98). MR–Egger regression gave similar results, suggesting that directional pleiotropy was unlikely to be biasing the result (intercept 0.002; p = 0.654). However, the apparent protective influence of higher BMI could be at least partially induced by survival bias in the PD GWA study, as demonstrated by frailty simulations. Other important limitations of this application of MR include the inability to analyse non-linear associations, to undertake subgroup analyses, and to gain mechanistic insights.

#### Conclusions

In this large study using two-sample MR, we found that variants known to influence BMI had effects on PD in a manner consistent with higher BMI leading to lower risk of PD. The mechanism underlying this apparent protective effect warrants further study.

## Author summary

## Why was this study done?

- There have been a number of conflicting observational studies describing associations between higher body mass index (BMI) and the future risk of Parkinson disease (PD). The most recent meta-analysis of cohort studies suggested that, overall, there was no effect.
- Observational studies have limitations, such that causal associations may not be reliably ascertained.
- This is the first time, to our knowledge, Mendelian randomisation has been used to study a potential causal association between genetically determined higher BMI and risk of PD.

## What did the researchers do and find?

- We used two-sample Mendelian randomisation, in which the relationship between genetic variants and BMI was ascertained from one study, and the relationship between the same genetic variants and PD was ascertained from a second study. Using this method, one can determine the estimated change in risk of PD per unit change in BMI.
- Our analysis indicated a potential causal influence of higher BMI protecting against PD.
- A limitation of the approach is that individuals who have higher BMI have a higher risk of earlier mortality, and therefore individuals with lower BMI may be over-represented amongst individuals diagnosed with PD. This "frailty effect" could at least partially account for the estimated causal effect.

Abbreviations: BMI, body mass index; GWA, genome-wide association; IPDGC, International Parkinson Disease Genomics Consortium; IV, instrumental variable; IVW, inverse-variance weighted; LD, linkage disequilibrium; MR, Mendelian randomisation; OR, odds ratio; PD, Parkinson disease; SE, standard error.

#### What do these findings mean?

- Further investigation, including exploring potential mechanisms, is warranted.
- Although our results suggest that higher BMI is potentially protective against PD, the negative health impacts of raising BMI are likely to be significant, and should be taken into account.

## Introduction

The risk of many disease outcomes increases as body mass index (BMI) increases, including cardiovascular disease and cancer [1]. Low BMI has also been associated with excess mortality (e.g., due to lung disease), but it is unknown whether these relationships are causal [1,2].

Parkinson disease (PD) is the second most common neurodegenerative disease worldwide [3]. The population burden is increasing for PD, relative to other diseases, even after population ageing is taken into account, and there is no cure [4]. A wide range of potential risk factors for PD have been described, mainly in observational studies [5,6].

The role of BMI in risk of PD is unclear, with observational studies providing conflicting results. A meta-analysis of case–control studies found significantly lower BMI in patients with PD compared with controls, but reverse causality could explain these findings (i.e., weight loss as a result of PD) [7]. A nested case–control study suggested that weight loss may precede the clinical diagnosis of PD [8]. Other studies have found negative associations between BMI and PD, some have found apparent null associations, and one has found a positive association [9–12].

Ten cohort studies were recently meta-analysed, and the pooled odds ratio (OR) for the association of a 5-kg/m<sup>2</sup> higher BMI with risk of PD was 1.00 (95% 0.89–1.12) [13]. However, there was substantial heterogeneity between studies ( $I^2 = 65\%$ ; p = 0.003), and individual studies may have been affected by residual confounding, bias, and, again, reverse causality (the latter can occur even in cohort studies, because PD has a long prodromal phase prior to diagnosis).

Mendelian randomisation (MR) is an instrumental variable (IV)–based method to infer causality in observational studies [14]. Gene variants that are associated with intermediate phenotypes or environmental exposures can be used as IVs to estimate the effect of the exposure on a disease outcome (see Fig 1). Random assortment of gene variants during gametogenesis means that potential confounding factors (observed and unobserved) are more likely to be evenly distributed, and the effect estimate that results from IV analysis is likely to be free from residual confounding and reverse causality. Given the difficulties in identifying causal risk factors for PD, MR has great potential to advance aetiological knowledge and identify putative therapeutic targets [15]. Methods are evolving rapidly as more in silico datasets become available in the form of extensive genetic data for a huge range of phenotypes and outcomes. Two-sample MR involves measuring variant–exposure associations in one dataset and variant–outcome associations in a second [16]. However, MR relies on certain assumptions:

1. The IV is robustly associated with the exposure of interest. This assumption can be checked by calculating an *F* statistic and  $R^2$  value. The IV may explain only a tiny amount of variance in the exposure ( $R^2$ ), and therefore studies often have to be large. As a result of genome-wide association (GWA) studies, there are increasing numbers of genetic variants



**Fig 1. Directed acyclic graph of instrumental variable analysis using genetic variants as proxies for environmental exposures (adapted from Lawlor et al. [14]).** Genetic variants (*Z*) associated with an exposure such as BMI (*X*) can be used as proxies to determine the effect of the exposure (*X*) on the outcome (*Y*). The three IV assumptions are indicated by arrows or the absence of arrows: (1) the IV in this schematic (*FTO* gene variant) is robustly associated with the exposure; (2) the IV is not associated with confounding factors (*C*); and (3) there is no alternative way that the IV affects the outcome other than via the exposure. BMI, body mass index; IV, instrumental variable; PD, Parkinson disease.

https://doi.org/10.1371/journal.pmed.1002314.g001

that account for the variance in a range of exposures and outcomes, allowing instruments of greater strength to be constructed.

- 2. **The IV is independent of known confounders.** In MR studies that have individual-level data, one can check for known confounders and compare the frequency of these between the two levels of the IV. In two-sample MR, the absence of individual-level data about potential confounders can hamper the ability to test this assumption.
- 3. The IV is independent of the outcome, given the exposure and confounders. In other words, there must be no alternative path, other than via the exposure, that the IV influences the outcome. This is known as the exclusion restriction criterion. One situation that violates this assumption is horizontal pleiotropy, in which there are alternative pathways through which the IV affects the outcome.

Here we describe the use of two-sample MR to estimate the causal association between genetically conferred variance in BMI and risk of PD.

#### Methods

Two-sample MR was undertaken using GWA study data. Ethical approval was not sought for this specific project because all data came from the summary statistics of published GWA studies, and no individual-level data were used.

#### Genetic variant instruments for body mass index

SNPs from the largest GWA study of BMI to date were identified from the 2015 summary statistic files from the GIANT (Genetic Investigation of Anthropometric Traits) consortium (http://portals.broadinstitute.org/collaboration/giant/index.php/GIANT\_consortium) [17]. Data on major and minor alleles for each SNP (after imputation, 2,554,637 variants in 339,224 individuals of European descent), along with allele frequencies, beta coefficients for allele dose and 5-kg/m<sup>2</sup> change in BMI (i.e., the change in BMI on a 5-kg/m<sup>2</sup> scale per effect allele), *p*-values, and standard errors (SEs) were extracted. This large number of variants was "clumped" to obtain a set of index SNPs that were independent of each other and associated with BMI at the genome-wide significance level (i.e., *p*-value  $< 5 \times 10^{-8}$ ). Index SNPs were identified by ranking BMI associations from the smallest to largest *p*-value (but still with a cutoff value of  $p = 5 \times 10^{-8}$ ). Clumped SNPs were those in linkage disequilibrium (LD) with index SNPs ( $R^2$  threshold of 0.001) or within 10,000 kb physical distance, based on a reference dataset (1000 Genomes Project; http://www. internationalgenome.org/). Hence, each index SNP represented a number of clumped SNPs that were all associated with or near to the index SNP, and the index SNPs were all independent of one another (according to the parameters defined here). Independence of index SNPs is important because bias can be introduced if there is LD between them and can result to over-precise estimates in subsequent analysis. Standard code for clumping is available on the PLINK website (http://zzz.bwh.harvard.edu/plink/clump.shtml), and further information is provided in S1 Appendix.

After clumping, there were 78 independent SNPs that were associated with BMI ( $p < 5 \times 10^{-8}$ ), and together these explained 2.2% of the variance in BMI ( $R^2 = 0.022$ ), as calculated from the summary files from the GIANT consortium [17]. To test the statistical significance of the association of the instrument with BMI, an *F* statistic was calculated using the following formula, where *k* is the number of variants and *n* is the sample size:

$$F = \frac{R^2(n-1-k)}{(1-R^2) \times k}$$

#### Association of body mass index genetic variants with Parkinson disease

PD genotyping data were from the most recent meta-analysis of GWA studies in PD, which related 7,782,514 genetic variants (after imputation) to PD, in up to 13,708 PD cases and 95,282 controls from 15 independent GWA datasets of individuals of European descent; the meta-analysis was undertaken by the International Parkinson Disease Genomics Consortium (IPDGC; http://pdgenetics.org/) [18]. We extracted the per-allele log-OR of PD together with its SE for each of the independent, genome-wide significant BMI SNPs identified from the GIANT consortium. It was possible to do this for 76 of the 78 BMI SNPs; for one SNP (rs1016287) a suitable proxy in high LD was available (rs887912;  $R^2 = 1.0$ ), but for another SNP (rs2245368), no suitable proxy (with  $R^2 > 0.9$ ) could be found, leaving 77 variants available for IV analysis. Three SNPs (rs1558902, rs17001654, and rs4256980) were palindromic, resulting in potential strand ambiguity. Allele frequencies for these were compared between the BMI and PD datasets to ensure that effect estimates were recorded with respect to the same effect allele.

#### Two-sample Mendelian randomisation methods

Two-sample MR was undertaken using previously described methods and as summarised below [19,20]. Wald ratios ( $\beta_{IV}$ ) were calculated for each of the 77 SNPs by dividing the perallele log-OR of PD ( $\beta_{ZY}$ ) by the per-allele difference in mean BMI for each SNP ( $\beta_{ZX}$ ):

$$\beta_{IV} = \beta_{ZY} / \beta_{ZX}$$

95% confidence intervals (95% CIs) were calculated from the SE of each Wald ratio, which was derived from the SE of the variant–outcome association divided by the variant–exposure association. Conventional linear regression analysis of the variant–exposure association and variant–outcome association for each instrument was undertaken and weighted by inverse

variance. The regression was constrained to pass through the origin, which forces the assumption that when the exposure has a value of zero, the outcome is also zero. This is known as the inverse-variance weighted (IVW) method, and the point estimate is equal to that derived from fixed-effect meta-analysis. The IVW method assumes that all variants are valid IVs.

Individual Wald ratios and 95% CIs were compiled in a forest plot. Heterogeneity in Wald ratios was tested using Cochran's Q and quantified using the  $I^2$  test. A "leave one out" sensitivity analysis was undertaken to identify variants with disproportionate effects, again using the IVW method.

To establish that violations of the third assumption of IV analysis were not biasing the estimate of the causal association (i.e., that there was not an aggregate unbalanced horizontal pleiotropic effect), MR–Egger regression was used. MR–Egger regression is similar to IVW regression, except that the intercept is not constrained to pass through the origin [20]. Substantial heterogeneity in the IVW estimate indicates that alternative pathways may exist from some of the SNPs to the outcome (known as horizontal pleiotropy), but this does not necessarily bias the estimate. However, a non-zero intercept from the MR–Egger regression suggests that pleiotropic effects tend to be in the direction of the intercept term, which will bias IVW estimates. A statistical hypothesis test can be performed to see if there is evidence of the intercept being different from zero, indicating overall unbalanced pleiotropy. The slope of the MR– Egger regression provides the estimate of the effect of BMI on PD when the third assumption is relaxed. The estimate is correct providing an additional assumption holds, the InSIDE (instrument strength independent of direct effect) assumption. This assumption states that the associations between genetic variants and the exposure are independent of the direct effects of the variants on the outcome [20].

In the absence of unbalanced pleiotropy, the IV estimates for individual SNPs ought to be symmetrically distributed around the point estimate, which, if centred on the IVW estimate, suggests that the result is not systematically biased. This can be demonstrated using a funnel plot of the individual variant effects plotted against the inverse of their SE.

We also repeated the IVW and MR–Egger analyses using the original 97 independent variants reported by the GIANT consortium (i.e., without the clumping step described above) [17]. Similar to the analyses with the 77 clumped variants, two variants were not available in the IPDGC data. The same proxy was used for one, and no proxy was found for the other, leaving 96 variants in the instrument.

Power calculations were undertaken using the proportion of variance in BMI explained by the 77 independent SNPs ( $R^2 = 0.022$ ) and the methods described by Brion and colleagues [21]. There was 92% power to detect a relative 20% difference in PD risk for a 5-kg/m<sup>2</sup> difference in BMI (i.e., an OR of at least 0.80 or 1.20) in the IPDGC cases and controls with an alpha of 5% (*p*-value  $\leq 0.05$ ). Power reduced to 38% to detect a relative difference in risk of 10%. These power calculations assumed no heterogeneity.

#### Frailty analysis

Studies of disorders that are strongly influenced by age (such as neurodegenerative disease) may be prone to bias if selective mortality has occurred [22]. For example, if people with high BMI died prematurely before being diagnosed with PD, then bias could occur because individuals with lower BMI live longer, resulting in a greater risk of being diagnosed with PD. Such an induced association would not reflect any biological link between BMI and PD. We performed simulations to estimate the likely effect that our MR analysis would show due to survival bias, assuming that BMI was not related to PD. The objective was to see if the likely magnitude of the survival bias was large enough to explain the MR results estimated from the real data. We performed simulations where a large sample (n = 500,000) was generated with data on BMI, SNPs influencing BMI, age, mortality status, and PD status. The variables were related using the following model:

- 1. Individuals were randomly assigned genotypes for each of 77 SNPs, with allele frequencies corresponding those in the MR analysis.
- 2. BMI values for each individual were simulated using their genotype values and the effect size (in standard deviation units) of each SNP, and a random value to create a population variance of one.
- 3. An age variable was generated for all individuals to match the distribution of ages in the PD meta-analysis.
- 4. Alive/dead status was simulated for individuals as a function of their age and BMI. Baseline age-related mortality rates and mortality rates for varying BMI levels were obtained from Davey Smith and colleagues [2]. These were used to generate a Gompertz–Makeham mortality curve, and individuals had their alive/dead status sampled as a function of the probability of death due to the survival curve.
- 5. PD status was simulated as a function of age-related diagnosis rates obtained from Driver et al. [23]; therefore, PD status was a function of age only, and unrelated to BMI level.
- 6. A subset of 13,708 simulated individuals with PD and 95,282 without PD, all of whom survived the mortality function, were then retained, based on the distribution of ages of individuals in the Nalls et al. PD GWA study [18]. Observational associations and MR were performed on these individuals to gauge the extent to which an association between BMI and PD was induced artificially by frailty effects.
- 7. The entire process was repeated 1,000 times to obtain a distribution of the effect size that was due to frailty effects only.

Full details about the simulations are available in <u>S1 Appendix</u>. All analyses were undertaken in R (version 3.2.3).

## Results

Table 1 shows the 77 variants selected for the analysis, with the effect alleles and frequencies, the magnitude of the effect on BMI, and strength of the association with PD. Further information on each variant, including regional genes and functions, are given in S1 and S2 Tables in S1 Appendix. The per-allele results of the IV analysis are provided in S3 Table in S1 Appendix. The *F* statistic for the instrument and its association with BMI was 99, which is large. This means that weak instrument bias was unlikely.

Using the IVW method to pool results from individual SNPs, genetically conferred higher BMI was found to be associated with a reduced risk of PD. The OR of PD per 5-kg/m<sup>2</sup> higher BMI was 0.82 (95% CI 0.69–0.98; p = 0.029). There was minimal evidence of heterogeneity between variants (*Q* statistic = 95.5;  $I^2 = 20.4\%$ ; p = 0.065; see Fig 2 and S3 Table in S1 Appendix). For the observed OR of 0.82, the estimated statistical power was 85% when the alpha was set at 5%.

Results from a "leave one out" analysis demonstrated that no single SNP was driving the IVW point estimate (see S4 Table in <u>S1 Appendix</u>). For example, after removal of the rs1558902 SNP, in the *FTO* locus, which has a strong magnitude of association with BMI, the OR of PD per 5-kg/m<sup>2</sup> higher BMI was 0.84 (95% CI 0.70–1.00).

able 1. Variants and effect alleles with frequencies and magnitude of effect on BMI and strength of association with PE	).

SNP	Chr	BP	EA	Non- EA	EA freq*	EA BMI beta	SE BMI beta	<i>p</i> -Value BMI association	EA PD log- OR	SE PD log- OR	<i>p</i> -Value PD association
rs17001654	4	77348592	G	С	0.154	0.030	0.005	7.76 × 10 <sup>-09</sup>	-0.077	0.024	0.0014
rs13107325	4	103407732	Т	С	0.072	0.047	0.007	1.83 × 10 <sup>-12</sup>	-0.109	0.032	0.0007
rs4787491	16	29922838	G	А	0.509	0.015	0.003	2.70 × 10 <sup>-08</sup>	-0.031	0.016	0.0508
rs4740619	9	15624326	Т	С	0.540	0.017	0.003	4.56 × 10 <sup>-09</sup>	-0.029	0.017	0.0777
rs2820292	1	200050910	С	А	0.547	0.018	0.003	1.83 × 10 <sup>-10</sup>	-0.031	0.016	0.0587
rs1808579	18	19358886	С	Т	0.534	0.016	0.003	4.17 × 10 <sup>-08</sup>	-0.027	0.016	0.0913
rs3888190	16	28796987	Α	С	0.397	0.031	0.003	3.14 × 10 <sup>-23</sup>	-0.045	0.017	0.0076
rs2033732	8	85242264	С	Т	0.749	0.018	0.003	4.89 × 10 <sup>-08</sup>	-0.023	0.019	0.2352
rs6465468	7	95007450	Т	G	0.300	0.016	0.003	4.98 × 10 <sup>-08</sup>	-0.020	0.019	0.2844
rs12401738	1	78219349	Α	G	0.347	0.020	0.003	1.15 × 10 <sup>-10</sup>	-0.023	0.017	0.1812
rs1000940	17	5223976	G	А	0.320	0.018	0.003	1.28 × 10 <sup>-08</sup>	-0.02	0.017	0.2477
rs6091540	20	50521269	С	Т	0.725	0.019	0.003	2.15 × 10 <sup>-11</sup>	-0.02	0.018	0.2659
rs11727676	4	145878514	Т	С	0.911	0.037	0.006	2.55 × 10 <sup>−08</sup>	-0.037	0.042	0.3745
rs16851483	3	142758126	Т	G	0.066	0.048	0.008	3.55 × 10 <sup>-10</sup>	-0.046	0.033	0.1647
rs10968576	9	28404339	G	А	0.315	0.025	0.003	6.61 × 10 <sup>-14</sup>	-0.024	0.017	0.1669
rs10132280	14	24998019	С	А	0.674	0.022	0.003	1.14 × 10 <sup>-11</sup>	-0.021	0.018	0.2513
rs492400	2	219057996	С	Т	0.426	0.015	0.003	6.78 × 10 <sup>-09</sup>	-0.014	0.017	0.4062
rs7899106	10	87400884	G	А	0.057	0.038	0.007	2.96 × 10 <sup>-08</sup>	-0.034	0.039	0.3811
rs1441264	13	78478920	Α	G	0.613	0.017	0.003	2.96 × 10 <sup>−08</sup>	-0.015	0.016	0.3631
rs3101336	1	72523773	С	Т	0.611	0.032	0.003	2.66 × 10 <sup>-26</sup>	-0.026	0.016	0.1203
rs13078960	3	85890280	G	Т	0.193	0.029	0.004	1.74 × 10 <sup>-14</sup>	-0.022	0.020	0.2624
rs17094222	10	102385430	С	Т	0.209	0.025	0.004	5.94 × 10 <sup>-11</sup>	-0.019	0.020	0.3432
rs11165643	1	96696685	Т	С	0.574	0.022	0.003	2.07 × 10 <sup>-12</sup>	-0.016	0.016	0.3277
rs12940622	17	76230166	G	А	0.572	0.018	0.003	2.49 × 10 <sup>-09</sup>	-0.012	0.016	0.4607
rs11030104	11	27641093	Α	G	0.791	0.042	0.004	5.56 × 10 <sup>-28</sup>	-0.027	0.020	0.1780
rs7599312	2	213121476	G	А	0.721	0.021	0.003	1.17 × 10 <sup>-10</sup>	-0.013	0.018	0.4813
rs6567160	18	55980115	С	Т	0.236	0.056	0.004	3.93 × 10 <sup>-53</sup>	-0.03	0.019	0.1144
rs2176040	2	226801046	Α	G	0.362	0.015	0.003	9.99 × 10 <sup>-09</sup>	-0.008	0.017	0.6443
rs1516725	3	187306698	С	Т	0.869	0.045	0.004	1.89 × 10 <sup>-22</sup>	-0.022	0.024	0.3480
rs17724992	19	18315825	Α	G	0.743	0.020	0.003	3.42 × 10 <sup>-08</sup>	-0.009	0.018	0.6166
rs7715256	5	153518086	G	Т	0.422	0.017	0.003	8.85 × 10 <sup>-09</sup>	-0.008	0.016	0.6363
rs12429545	13	53000207	Α	G	0.135	0.032	0.004	1.09 × 10 <sup>-12</sup>	-0.011	0.024	0.6440
rs1558902	16	52361075	Α	Т	0.409	0.081	0.003	7.51 × 10 <sup>-153</sup>	-0.022	0.017	0.1859
rs17203016	2	207963763	G	A	0.195	0.021	0.004	3.41 × 10 <sup>-08</sup>	-0.006	0.021	0.7884
rs657452	1	49362434	Α	G	0.397	0.023	0.003	5.48 × 10 <sup>-13</sup>	-0.006	0.017	0.7212
rs1460676	2	164275935	С	Т	0.179	0.021	0.004	4.98 × 10 <sup>-08</sup>	-0.005	0.022	0.8204
rs6804842	3	25081441	G	A	0.569	0.018	0.003	2.48 × 10 <sup>-09</sup>	-0.004	0.016	0.7945
rs4256980	11	8630515	G	С	0.638	0.021	0.003	2.90 × 10 <sup>-11</sup>	-0.005	0.017	0.7876
rs1528435	15	65864222	Т	С	0.631	0.018	0.003	1.20 × 10 <sup>-08</sup>	-0.004	0.017	0.8292
rs16951275	2	181259207	Т	С	0.771	0.030	0.004	1.91 × 10 <sup>−17</sup>	-0.006	0.019	0.7585
rs10182181	2	25003800	G	A	0.468	0.031	0.003	8.78 × 10 <sup>-24</sup>	-0.006	0.016	0.7275
rs17024393	1	109956211	С	Т	0.043	0.061	0.008	7.03 × 10 <sup>-14</sup>	-0.01	0.049	0.8327
rs3817334	11	47607569	Т	С	0.401	0.026	0.003	5.15 × 10 <sup>−17</sup>	-0.004	0.016	0.8003
rs1928295	9	119418304	Т	С	0.550	0.018	0.003	7.91 × 10 <sup>-10</sup>	-0.003	0.016	0.8747
rs2112347	5	75050998	Т	G	0.621	0.025	0.003	6.19 × 10 <sup>-17</sup>	-0.003	0.022	0.9055
rs2365389	3	61211502	С	Т	0.572	0.020	0.003	1.63 × 10 <sup>-10</sup>	-0.001	0.017	0.9654
rs758747	16	3567359	Т	С	0.280	0.023	0.004	7.47 × 10 <sup>-10</sup>	0.000	0.018	0.9996

(Continued)

#### Table 1. (Continued)

SNP	Chr	BP	EA	Non- EA	EA freq*	EA BMI beta	SE BMI beta	<i>p</i> -Value BMI association	EA PD log- OR	SE PD log- OR	<i>p</i> -Value PD association
rs13201877	6	137717234	G	A	0.140	0.024	0.004	4.29 × 10 <sup>-08</sup>	0.003	0.026	0.9138
rs7138803	12	48533735	Α	G	0.379	0.032	0.003	8.15 × 10 <sup>-24</sup>	0.005	0.017	0.7605
rs2287019	19	50894012	С	Т	0.806	0.035	0.004	4.59 × 10 <sup>-18</sup>	0.007	0.020	0.7198
rs2207139	6	50953449	G	А	0.176	0.045	0.004	4.13 × 10 <sup>-29</sup>	0.011	0.021	0.6064
rs17405819	8	76969139	Т	С	0.702	0.022	0.003	2.07 × 10 <sup>-11</sup>	0.006	0.017	0.7158
rs10938397	4	44877284	G	А	0.428	0.040	0.003	3.21 × 10 <sup>-38</sup>	0.014	0.016	0.4092
rs29941	14	78969207	G	А	0.670	0.018	0.003	2.41 × 10 <sup>-08</sup>	0.006	0.017	0.7122
rs7141420	19	39001372	Т	С	0.529	0.023	0.003	1.23 × 10 <sup>-14</sup>	0.008	0.017	0.6259
rs2033529	6	40456631	G	А	0.289	0.018	0.003	1.39 × 10 <sup>-08</sup>	0.006	0.018	0.7176
rs6477694	9	110972163	С	Т	0.371	0.017	0.003	2.67 × 10 <sup>-08</sup>	0.006	0.017	0.7016
rs13021737	2	622348	G	A	0.830	0.060	0.004	1.11 × 10 <sup>-50</sup>	0.023	0.021	0.2741
rs1167827	7	75001105	G	А	0.557	0.020	0.003	6.33 × 10 <sup>-10</sup>	0.009	0.016	0.6014
rs7903146	10	114748339	С	Т	0.713	0.024	0.003	1.11 × 10 <sup>-11</sup>	0.011	0.018	0.5310
rs543874	1	176156103	G	A	0.195	0.050	0.004	2.62 × 10 <sup>-35</sup>	0.025	0.020	0.2265
rs205262	6	34671142	G	А	0.285	0.021	0.003	1.75 × 10 <sup>-10</sup>	0.011	0.018	0.5442
rs11057405	12	121347850	G	A	0.902	0.030	0.005	2.02 × 10 <sup>-08</sup>	0.018	0.030	0.5438
rs7239883	18	38401669	G	A	0.394	0.015	0.003	1.51 × 10 <sup>-08</sup>	0.01	0.016	0.5618
rs2836754	21	39213610	С	Т	0.599	0.017	0.003	1.61 × 10 <sup>-08</sup>	0.011	0.017	0.5139
rs12286929	11	114527614	G	A	0.523	0.021	0.003	1.31 × 10 <sup>-12</sup>	0.014	0.016	0.3858
rs2176598	11	43820854	Т	С	0.256	0.019	0.003	2.97 × 10 <sup>-08</sup>	0.013	0.018	0.4701
rs3736485	15	49535902	Α	G	0.461	0.016	0.003	7.41 × 10 <sup>-09</sup>	0.013	0.016	0.4350
rs10733682	9	128500735	Α	G	0.475	0.019	0.003	1.83 × 10 <sup>-08</sup>	0.017	0.016	0.2755
rs13191362	6	162953340	Α	G	0.880	0.029	0.005	7.34 × 10 <sup>-09</sup>	0.035	0.025	0.1626
rs9374842	6	120227364	Т	С	0.748	0.020	0.003	2.67 × 10 <sup>-08</sup>	0.026	0.019	0.1726
rs887912	2	59075742	Т	С	0.317	0.023	0.003	2.75 × 10 <sup>-11</sup>	0.031	0.018	0.0738
rs9540493	13	65103705	Α	G	0.464	0.018	0.003	4.97 × 10 <sup>-08</sup>	0.025	0.017	0.1389
rs977747	1	47457264	С	G	0.403	0.017	0.003	2.18 × 10 <sup>-08</sup>	0.024	0.016	0.1368
rs2121279	2	142759755	Т	С	0.150	0.024	0.004	2.31 × 10 <sup>-08</sup>	0.037	0.024	0.1217
rs9400239	6	109084356	С	Т	0.673	0.017	0.003	1.61 × 10 <sup>-08</sup>	0.028	0.018	0.1180
rs3849570	3	81874802	Α	С	0.362	0.018	0.003	2.60 × 10 <sup>-08</sup>	0.030	0.017	0.0815

#### \*Frequency of effect allele in Locke et al. [17].

BMI, body mass index; BP, base pair position; Chr, chromosome; EA, effect allele; OR, odds ratio; PD, Parkinson disease; SE, standard error.

https://doi.org/10.1371/journal.pmed.1002314.t001

The effect estimated from MR–Egger regression was 0.76 (95% CI 0.51–1.14; p = 0.177), with an intercept of 0.002 (95% CI –0.008 to 0.013; p = 0.654). A funnel plot (see S1 Fig in S1 Appendix) suggested that individual variants were symmetrically distributed around the point estimate. Together these findings provide evidence against the possibility that systematic bias in the IVW estimate may have arisen through overall unbalanced horizontal pleiotropy.

We repeated the analysis using the original set of "un-clumped" variants reported by the GIANT consortium (96 variants in total) [17]. This reanalysis did not alter the observed effect of genetically estimated BMI on risk of PD (IVW OR 0.80, 95% CI 0.67–0.97; see S2 Fig in <u>S1</u> Appendix).

We performed simulations to obtain estimates of effect sizes that would be induced by survival bias alone, under the null hypothesis that BMI is not biologically related to PD. If the effect sizes obtained from these simulations were similar in magnitude to the effect sizes we



Odds ratio & 95% confidence interval

Fig 2. Forest plot of Wald ratios and 95% Cls generated from clumped SNPs associated with BMI. Odds ratios for individual SNPs are listed according to magnitude of effect in the instrumental variable analysis and are presented with pooled

effects using the IVW method and MR–Egger regression. The most recent meta-analysis of observational studies is also plotted (Wang et al. [13]). Squares represent the point estimate, and the bars are the 95% confidence intervals. There was weak evidence of heterogeneity (*Q* statistic = 95.5;  $\hat{F} = 20.4\%$ ; p = 0.065). IVW, inverse-variance weighted; MR, Mendelian randomisation.

https://doi.org/10.1371/journal.pmed.1002314.g002

estimated from the real data, it would suggest that survival bias was sufficient to explain the results. Simulations using empirical BMI- and age-related mortality rates demonstrated that a likely influence of frailty on biasing the MR analysis of BMI and risk of PD was present, but smaller than the estimated inverse influence of BMI on PD from the empirical MR analysis (see Fig 3). The mean induced effect due to frailty alone in the IVW analysis was an OR of 0.97 per 5-kg/m<sup>2</sup> higher BMI (95% CI 0.92–1.03), compared to our empirical estimate of the OR of PD per 5-kg/m<sup>2</sup> higher BMI of 0.82 (95% CI 0.69–0.98).



Odds ratio per 5 kg/m2 and 95% confidence intervals

**Fig 3. Frailty analysis.** We obtained effect estimates from two sources: (1) analysis of the causal influence of BMI on PD using real data and (2) simulations where BMI had no effect on PD, and any apparent effect was due to survival bias alone. This figure shows the comparison of the estimates from these two sources using three different approaches—MR IVW analysis, MR–Egger regression, and observational associations. For the true effect estimates, the horizontal lines denote the 95% confidence intervals; for the results from simulations, the horizontal lines denote 95% confidence intervals. BMI, body mass index; IVW, inverse-variance weighted; MR, Mendelian randomisation; PD, Parkinson disease.

https://doi.org/10.1371/journal.pmed.1002314.g003

## Discussion

This study provides MR evidence that higher BMI could protect against the risk of developing PD. Simulation studies performed to ascertain the effect of survivor bias showed that survivor bias could be a contributing factor but did not explain all of the effect. To our knowledge, this is the first MR study of this association and was undertaken in a very large sample. Genetic elements expected to lead to a lifetime elevation in BMI of 5 kg/m<sup>2</sup> were associated with a lower risk of PD, indicating a causal influence of BMI on PD (OR 0.82). Whilst there was minimal evidence of heterogeneity across individual SNP estimates, the MR–Egger analysis supported our main study findings, with an intercept value close to zero and a point estimate for the effect that was similar to that of the main analysis (see S3 Fig in S1 Appendix). MR–Egger regression has less statistical power than an equivalent IVW meta-analysis, and hence the confidence intervals for this analysis were wider and included the null value.

These findings shed additional light on an issue for which the evidence to date has been mixed. Prior to PD diagnosis, observational studies have found conflicting results with regard to the association between BMI and risk of PD [8–12]. The conflicting results may be due to study-specific biases, differential adjustment for confounding factors, and use of different cut-offs to define the exposure (see Table 2). A recent meta-analysis of these and additional studies concluded that there was no robust observational association [13]. There was substantial heterogeneity between the effect estimates of individual studies ( $I^2 = 65\%$ ; p = 0.003), raising the

Authors and year [reference]	Study design and location	Observed association	Handling of potential sources of bias	Handling of confounding		
Ma et al. 2006 [8]	Nested case–control study in Linxian County, China. Risk of reverse causality reduced because measurement of BMI preceded ascertainment of the outcome.	BMI inversely associated with risk of PD, with dose- dependent effect for increasing BMI.	Two-step ascertainment of PD diagnosis and standard clinical criteria to reduce bias. Only definite cases included. Generalisability may be limited by cohort and location.	Cases matched to controls 1:4 by age and sex. Multiple logistic regression analysis modelling BMI, gastric ulcers, meat consumption, and pack-years smoking (all associated with PD in univariate analysis).		
Chen et al. 2004 [9]	Health Professionals Follow-up Study and Nurses' Health Study cohorts, US; excluded cases diagnosed within 3–4 y of baseline.	Baseline BMI not associated with risk of PD, but most recently recorded BMI was inversely associated with PD.	Questionnaire during follow-up included item about diagnosis of PD, with positive responses confirmed by neurological examination. This approach misses undiagnosed cases and biases estimates towards the null value. Generalisability may be limited by cohort composition.	Associations adjusted for age, smoking, caffeine intake, and alcohol intake. Men and women presented separately and in pooled analysis.		
Logroscino et al. 2007 [11]	Harvard Alumni Health Study cohort, US; 63,557 person-years of follow-up; excluded cases diagnosed within 4 y of baseline.	Trend for inverse association between BMI measured in 1988 and PD. No association between BMI measured in college and PD.	Self-reported diagnosis of PD (previous validation study found 70% to be correct). Non-differential misclassification of the outcome would bias estimates towards null.	Associations adjusted for age, smoking, caffeine intake, physical activity, and previous cardiovascular disease or cancer.		
Hu et al. 2006 [12]	Six independent population surveys, Finland. Mean follow-up 18.8 y; excluded cases diagnosed within 5 y of baseline.	BMI positively associated with PD, with dose- dependent effect for increasing BMI.	PD diagnosis from national drug reimbursement register. May under- ascertain cases with low BMI— differential misclassification—if obese individuals are more likely to contact health system.	Associations adjusted for age, study year, blood pressure, cholesterol level, education, physical activity, smoking, caffeine intake, and alcohol intake. Men and women presented separately and booled.		

Table 2. Examples of previous observational studies of body mass index and Parkinson disease.

#### BMI, body mass index; PD, Parkinson disease.

https://doi.org/10.1371/journal.pmed.1002314.t002

question as to whether it was appropriate to perform meta-analysis. Three of these studies did not set out specifically to investigate the effect of BMI on the risk of PD [8,24,25]. Two of the studies showed a clear negative association between higher BMI and the risk of PD [8,11]. Many of the studies used self-report questionnaires to ascertain PD, which is likely to have resulted in under-ascertainment of the outcome and bias towards the null. These factors could have had a strong influence on the overall results of the meta-analysis.

Recently, a similar direction and magnitude of effect of higher BMI on risk of PD was reported from the United Kingdom Clinical Practice Research Datalink. In a cohort of nearly 2 million, underweight people had a 15% excess risk of PD compared with normal weight people, and overweight and obese people had a 12% and 17% reduction in risk, respectively [10].

The mechanism through which higher BMI may reduce the risk of PD is not immediately apparent, but researchers have recently reported on possible neuroprotective benefits arising in those with higher BMI, particularly relating to preservation of cognitive function and neural networks, explored using functional imaging [26]. Higher BMI affects levels of circulating and central insulin, which in turn may play a beneficial role with respect to neurodegeneration [27]. Theories such as these require substantial further exploration.

Weight loss is widely recognised in patients with a diagnosis of PD [7]. The mechanisms underlying body weight reduction in this context are easier to understand and are multifactorial, including intrinsic factors of the disease as well as disruption of both peripheral and central regulatory mechanisms of body weight, feeding behaviour, and energy metabolism [28]. Vitally, however, the mechanisms that drive risk of PD may not be the same as those that underlie progression once disease has occurred, and the MR study presented here is designed to investigate only the former.

There are a number of candidate genes (see S1 and S2 Tables in S1 Appendix) that are located in close proximity to SNPs included as IVs, which is intriguing and raises the possibility of pleiotropic pathways (i.e., not via BMI). *BDNF*, coding for Brain-Derived Neurotrophic Factor, is a gene that has previously been implicated in PD, and BDNF has been discussed as a potential therapeutic agent in neurodegenerative disease but has previously been unsuccessful in clinical trials in patients with amyotrophic lateral sclerosis [29–31]. *PARK2*, coding for the Ubiquitin E3 ligase Parkin, is mutated in rare autosomal recessive juvenile-onset parkinsonism cases, and is notable for not having been identified in a GWA study for PD [18,32]. Finally, the *RPTOR* gene (encoding Raptor, a regulator of the kinase mTOR) is of interest given the links between the mTOR pathway, macroautophagy, and PD [33]. For each of these examples, proof that these genes are directly linked to causation at these loci requires further research.

The finding of higher BMI reducing the risk of PD seemingly conflicts with the observation from a cohort study that physical activity may be a protective factor for PD [34]. Again, cohort studies such as this may be biased by confounding or reverse causality, and the association with physical activity may be driven simply by the fact that people with undiagnosed PD may undertake less activity as a result of occult disease. However, evidence of an effect of physical activity on BMI may be weaker than one might intuitively expect [35,36]. There may be true inverse causal associations for both BMI and physical activity with risk of PD, and physical activity may protect primarily through mechanisms that are independent of BMI. An MR study (or randomised controlled trial) of physical activity and risk of PD could be used to explore this further, if robust genetic variants could be identified to account for variability in this behaviour.

An MR study to determine an association between genetically conferred variance in BMI (alongside other potentially modifiable risk factors) and Alzheimer disease was recently reported [37]. The authors observed no similar protective effect to that which we observed for PD, but their instrument used only 32 variants. They did observe a reduction in risk of

Alzheimer disease for increasing blood pressure, but did not undertake a similar frailty analysis to that which we conducted.

Our frailty simulations sought to explore the extent to which our MR results could be explained by frailty effects alone. The influence of frailty effects should be an important consideration in epidemiological studies dealing with later life events [22]. We found that the effect of BMI on mortality and the age disparity between the cases and controls in the PD GWA study was unlikely to be sufficient to explain our MR estimate, particularly given that PD has a relatively early onset compared to some other neurological disorders (median age 60 y old) [3].

#### Strengths and weaknesses

MR uses an IV approach to assess causal relationships between environmental exposures/ intermediate phenotypes and disease outcomes, while minimising or eliminating the possibility of bias due to residual confounding or reverse causality. One key assumption is that the instrument affects PD risk through its effect on a specific phenotype/exposure (here this was BMI), and does not have a direct effect on PD risk independently of this (the exclusion restriction criterion, or assumption number three from the Introduction). We tested this assumption using MR–Egger regression and found no evidence of violation of this assumption.

This analysis was undertaken using summary statistics from the two largest datasets for BMI and PD. The use of different datasets to ascertain variant–exposure and variant–outcome associations, so-called two-sample MR, has methodological advantages over analysis undertaken in a single dataset, but also some limitations [38]. First, the data are available without having to undertake measurement of each association in turn, saving both time and cost. Second, statistical power is high because very large sample sizes can be achieved using this method. Third, if bias did arise in the analysis due to weak instruments, it would tend to be biased towards the null (i.e., give rise to conservative estimates), whereas in one-sample MR the estimate tends to be biased towards the confounded observational study estimate [16]. The first assumption of MR is that the instrument under study is strongly associated with the exposure. Here, the first-stage *F* statistic was large (99), and so weak instrument bias is unlikely. Furthermore, samples are assumed to be completely independent. The cohorts contributing to both the GIANT and IPDGC consortia are listed in S5 and S6 Tables in S1 Appendix. If there was some overlap in the controls from the two consortia, then an *F* statistic of this magnitude would mitigate potential bias.

Our frailty analysis used demographic information to model the induced bias due to the combination of BMI influencing mortality and cases being on average younger than controls in the PD GWA study. Alternative mechanisms of frailty may operate. For example, because BMI reduces longevity, individuals with high BMI who survive longer potentially over-represent unmeasured factors that increase longevity. If these factors, in turn, also influence risk of PD, then bias could be introduced into our MR estimates. This mechanism can influence observational associations also.

One weakness of our approach is that two-sample MR using aggregate data (as done here) does not currently allow one to examine non-linear relationships between exposures and outcomes. The association between BMI and most health-related outcomes tends to be U- or J-shaped, with very low weight and above average weight both resulting in adverse outcomes [1]. Here, in the MR analysis, BMI was modelled as a continuous variable, and linearity assumed. The use of aggregate data also means that we were not able to test differences of effect in subgroups. There is evidence to suggest that the causal effect of BMI on some outcomes, including socioeconomic status, may differ between men and women [39]. Using this design,

we cannot test whether effects of BMI on PD differ by sex. In addition, lack of information about potential confounding factors for the relationship of BMI and PD means that associations between the variants and smoking/alcohol behaviours could not be tested (assumption number two from the Introduction). Another limitation of this study is that it does not offer insight into the mechanisms by which BMI is causally related to PD. Finally, two-sample MR assumes that both samples come from the same population, but with no overlap. If they come from separate populations, the magnitude of the estimated causal effect could be biased. We took steps to ensure that both samples were of individuals of European ancestry.

In this large two-sample MR study exploring the association between genetically determined higher BMI and risk of PD, we observed a possible protective effect exerted by a potentially modifiable factor. MR should be seen as one tool to explore causal questions, but not as a definitive answer. However, robust empirical evidence to support a protective role of higher BMI in risk of PD may be hard to obtain. It is plausible that our finding could be replicated in large cohort studies such as the UK Biobank, which gathered information on a wide range of exposures and disease outcomes, along with genotyping data, on more than half a million older people in the UK [40]. Our study offers the best evidence to date that higher BMI may convey relative protection against PD, in support of clinical observation and some of the preexisting observational studies. However, the apparent protective influence of higher BMI could be at least partially induced by survival bias in the PD GWA study, as demonstrated in the frailty simulations.

Although BMI is a potentially modifiable risk factor for PD, the negative health impacts of raising BMI are likely to be significant, and should be taken into account. Expected negative effects would include increased risk of type 2 diabetes mellitus, ischaemic heart disease, and cancer. Causal associations have been demonstrated between higher BMI and many cancers, and have been explored for BMI and cardiovascular disease [41,42]. Assuming replication of a negative association between BMI and PD, a great deal of further work would be required (including mechanistic insights) before recommending this as a potential intervention against PD, given the wider public health effects.

## **Supporting information**

**S1 Appendix. Supplementary materials.** (DOCX)

## Acknowledgments

For full membership of the International Parkinson Disease Genomics Consortium, please see S1 Appendix.

We would like to thank all of the participants who donated their time and biological samples to be a part of this study. This work was supported in part by the Intramural Research Program of the National Institute of Neurological Disorders and Stroke, the National Institute on Aging, and the National Institute of Environmental Health Sciences, which are part of the National Institutes of Health, US Department of Health and Human Services, project numbers Z01-AG000949-02 and Z01-ES101986. In addition, this work was supported by the US Department of Defense (award W81XWH-09-2-0128), and the Michael J. Fox Foundation for Parkinson's Research. This work was supported by National Institutes of Health grants R01NS037167, R01CA141668, and P50NS071674; American Parkinson Disease Association; Barnes–Jewish Hospital Foundation; Greater St. Louis Chapter of the American Parkinson Disease Association; Hersenstichting Nederland; Neuroscience Campus Amsterdam; and the Section of Medical Genomics, the Prinses Beatrix Fonds. The KORA (Cooperative Health Research in the Augsburg Region) research platform was started and financed by the Forschungszentrum für Umwelt und Gesundheit, which is funded by the German Federal Ministry of Education and Research and by the State of Bavaria. This study was also funded by the German National Genome Research Network (NGFN-Plus number 01GS08134, German Federal Ministry of Education and Research), the German Federal Ministry of Education and Research (NGFN 01GR0468, PopGen), and 01EW0908 in the frame of ERA-NET NEURON and Helmholtz Alliance for Mental Health in an Ageing Society (HA-215), which was funded by the Initiative and Networking Fund of the Helmholtz Association. The French GWA study work was supported by the French National Research Agency (ANR-08-MNP-012). This study was also funded by Association France Parkinson, the French programme "Investissements d'avenir" funding (ANR-10-IAIHU-06), and a grant from Assistance Publique-Hôpitaux de Paris (PHRC AOR-08010) for the French clinical data. This study was also sponsored by the Landspitali University Hospital Research Fund, Icelandic Research Council, European Commission Framework Programme 7, People Programme, and IAPP on Novel Genetic and Phenotypic Markers of Parkinson's Disease and Essential Tremor (MarkMD), contract number PIAP-GA-2008-230596 MarkMD. This study utilised the high-performance computational capabilities of the Biowulf Linux cluster at the National Institutes of Health, Bethesda, Maryland, and DNA panels, samples, and clinical data from the National Institute of Neurological Disorders and Stroke Human Genetics Resource Center DNA and Cell Line Repository. People who contributed samples are acknowledged in descriptions of every panel on the repository website. We thank the French Parkinson's Disease Genetics Study Group and the Drug Interaction with Genes in Parkinson's Disease study group: Y. Agid, M. Anheim, A.-M. Bonnet, M. Borg, A. Brice, E. Broussolle, J.-C. Corvol, P. Damier, A. Destée, A. Dürr, F. Durif, A. Elbaz, D. Grabil, S. Klebe, P. Krack, E. Lohmann, L. Lacomblez, M. Martinez, V. Mesnage, P. Pollak, O. Rascol, F. Tison, C. Tranchant, M. Vérin, F. Viallet, and M. Vidailhet. We also thank the members of the French 3C Consortium (A. Alpérovitch, C. Berr, C. Tzourio, and P. Amouyel) for allowing us to use part of the 3C cohort, and D. Zelenika for support in generating the genome-wide molecular data. We thank P. Tienari (Molecular Neurology Research Programme, Biomedicum, University of Helsinki), T. Peuralinna (Department of Neurology, Helsinki University Central Hospital), L. Myllykangas (Folkhälsan Institute of Genetics and Department of Pathology, University of Helsinki), and R. Sulkava (Department of Public Health and General Practice, Division of Geriatrics, University of Eastern Finland) for the Finnish controls (Vantaa85+ GWA study data). We used GWA data generated by the Wellcome Trust Case Control Consortium 2 (WTCCC2) from UK patients with PD and UK control individuals from the 1958 Birth Cohort Study and National Blood Service. Genotyping of UK replication cases on ImmunoChip was part of the WTCCC2 project, which was funded by the Wellcome Trust (083948/Z/07/Z). UK population control data were made available through WTCCC1. This study was supported by the Medical Research Council and Wellcome Trust Disease Centre. As with previous IPDGC efforts, this study makes use of data generated by the Wellcome Trust Case Control Consortium. A full list of the investigators who contributed to the generation of the data is available from https://www.wtccc.org.uk/. Funding for the project was provided by the Wellcome Trust under awards 076113, 085475, and 090355. This study was also supported by Parkinson's UK (grants 8047 and J-0804) and the Medical Research Council (G0700943 and G1100643). We thank Jeffrey Barrett and Jason Downing for assistance with the design of the ImmunoChip and NeuroX arrays. DNA extraction work that was done in the UK was undertaken at University College London Hospitals, University College London, which received a proportion of funding from the Department of Health's National Institute for Health Research Biomedical Research Centre funding. This study was

supported in part by the Wellcome Trust/Medical Research Council Joint Call in Neurodegeneration award (WT089698) to the UK Parkinson's Disease Consortium, whose members are from the UCL Institute of Neurology, the University of Sheffield, and the Medical Research Council Protein Phosphorylation and Ubiquitylation Unit at the University of Dundee. Mike Nalls's participation is supported by a consulting contract between Kelly Services and the National Institute on Aging, National Institutes of Health, Bethesda, Maryland, US.

#### **Author Contributions**

Conceptualization: AJN, DAK, JH, ASc, AJL, ASi, TF, MAN, NP, DAL, NWW.

Data curation: AJN, DAK, GH, AN, TRP, EPF, PAL, GDS, DAL, MAN.

Formal analysis: AJN, DAK, GH, MAN.

Investigation: AJN, DAK, GH, MAN.

Methodology: AJN, DAK, GH, PCH, GDS, MAN, DAL, NP, NWW.

Software: GH, AN, TRP, MAN.

Supervision: DAL, MAN, NWW.

Validation: AJN, DAK.

Writing - original draft: AJN, DAK, GH, GDS, DAL, NP, MAN, NWW.

Writing - review & editing: AN, TRP, PCH, EPF, PAL, TF, ASc, AJL, ASi, PAL.

#### References

- Prospective Studies Collaboration. Body-mass index and cause-specific mortality in 900 000 adults: collaborative analyses of 57 prospective studies. Lancet. 2009; 373:1083–96. https://doi.org/10.1016/ S0140-6736(09)60318-4 PMID: 19299006
- Davey Smith G, Sterne JAC, Fraser A, Tynelius P, Lawlor DA, Rasmussen F. The association between BMI and mortality using offspring BMI as an indicator of own BMI: large intergenerational mortality study. BMJ. 2009; 339:b5043. https://doi.org/10.1136/bmj.b5043 PMID: 20028778
- Lees AJ, Hardy J, Revesz T. Parkinson's disease. Lancet. 2009; 373:2055–66. https://doi.org/10.1016/ S0140-6736(09)60492-X PMID: 19524782
- GBD 2013 Mortality and Causes of Death Collaborators. Global, regional, and national age-sex specific all-cause and cause-specific mortality for 240 causes of death, 1990–2013: a systematic analysis for the Global Burden of Disease Study 2013. Lancet. 2015; 385:117–71. https://doi.org/10.1016/S0140-6736(14)61682-2 PMID: 25530442
- Noyce AJ, Bestwick JP, Silveira-Moriyama L, Hawkes CH, Giovannoni G, Lees AJ, et al. Meta-analysis of early nonmotor features and risk factors for Parkinson disease. Ann Neurol. 2012; 72:893–901. https://doi.org/10.1002/ana.23687 PMID: 23071076
- 6. Noyce AJ, Lees AJ, Schrag A-E. The prediagnostic phase of Parkinson's disease. J Neurol Neurosurg Psychiatry. 2016; 87:871–8. https://doi.org/10.1136/jnnp-2015-311890 PMID: 26848171
- van der Marck MA, Dicke HC, Uc EY, Kentin UC, Borm GF, Bloem BR, et al. Body mass index in Parkinson's disease: a meta-analysis. Parkinsonism Relat Disord. 2012; 18:263–7. https://doi.org/10.1016/j. parkreldis.2011.10.016 PMID: 22100523
- Ma L, Zhang L, Gao XH, Chen W, Wu YP, Wang Y, et al. Dietary factors and smoking as risk factors for PD in a rural population in China: a nested case-control study. Acta Neurol Scand. 2006; 113:278–81. https://doi.org/10.1111/j.1600-0404.2005.00571.x PMID: 16542169
- Chen H, Zhang SM, Schwarzschild MA, Hernán MA, Willett WC, Ascherio A. Obesity and the risk of Parkinson's disease. Am J Epidemiol. 2004; 159:547–55. PMID: 15003958
- Qizilbash N, Gregson J, Johnson ME, Pearce N, Douglas I, Wing K, et al. Body mass index and risk of Parkinson's disease in a cohort of two million people over two decades. Pharmacoepiol Drug Saf. 2016; 25(Suppl 3):105.

- Logroscino G, Sesso HD, Paffenbarger RS, Lee I-M. Body mass index and risk of Parkinson's disease: a prospective cohort study. Am J Epidemiol. 2007; 166:1186–90. https://doi.org/10.1093/aje/kwm211 PMID: 17709328
- Hu G, Jousilahti P, Nissinen A, Antikainen R, Kivipelto M, Tuomilehto J. Body mass index and the risk of Parkinson disease. Neurology. 2006; 67:1955–9. https://doi.org/10.1212/01.wnl.0000247052.18422.e5 PMID: 17159100
- Wang Y-L, Wang Y-T, Li J-F, Zhang Y-Z, Yin H-L, Han B. Body mass index and risk of Parkinson's disease: a dose-response meta-analysis of prospective studies. PLoS ONE. 2015;10(6):e0131778. <a href="https://doi.org/10.1371/journal.pone.0131778">https://doi.org/10.1371/journal.pone.0131778</a> PMID: 26121579
- Lawlor DA, Harbord RM, Sterne JAC, Timpson N, Davey Smith G. Mendelian randomization: using genes as instruments for making causal inferences in epidemiology. Stat Med. 2008; 27:1133–63. https://doi.org/10.1002/sim.3034 PMID: 17886233
- Noyce AJ, Nalls MA. Mendelian randomization—the key to understanding aspects of Parkinson's disease causation? Mov Disord. 2015; 31:478–83. https://doi.org/10.1002/mds.26492 PMID: 26695521
- Pierce BL, Burgess S. Efficient design for Mendelian randomization studies: subsample and 2-sample instrumental variable estimators. Am J Epidemiol. 2013; 178:1177–84. <u>https://doi.org/10.1093/aje/</u> kwt084 PMID: 23863760
- Locke AE, Kahali B, Berndt SI, Justice AE, Pers TH, Day FR, et al. Genetic studies of body mass index yield new insights for obesity biology. Nature. 2015; 518:197–206. <u>https://doi.org/10.1038/nature14177</u> PMID: 25673413
- Nalls MA, Pankratz N, Lill CM, Do CB, Hernandez DG, Saad M, et al. Large-scale meta-analysis of genome-wide association data identifies six new risk loci for Parkinson's disease. Nat Genet. 2014; 46:989–93. https://doi.org/10.1038/ng.3043 PMID: 25064009
- Johnson T. Efficient calculation for multi-SNP genetic risk scores. Vienna: Comprehensive R Archive Network; 2012 [cited 2016 Aug 1]. Available from: http://cran.r-project.org/web/packages/gtx/vignettes/ ashg2012.pdf.
- Bowden J, Davey Smith G, Burgess S. Mendelian randomization with invalid instruments: effect estimation and bias detection through Egger regression. Int J Epidemiol. 2015; 44:512–25. <u>https://doi.org/10.1093/ije/dyv080 PMID: 26050253</u>
- Brion MJA, Shakhbazov K, Visscher PM. Calculating statistical power in Mendelian randomization studies. Int J Epidemiol. 2013; 42:1497–501. https://doi.org/10.1093/ije/dyt179 PMID: 24159078
- Mayeda ER, Tchetgen EJ, Power MC, Weuve J, Jacqmin-Gadda H, Marden JR, et al. A simulation platform for quanitifying survival bias: an application to research on determinants of cognitive decline. Am J Epidemiol. 2016; 184:378–87. https://doi.org/10.1093/aje/kwv451 PMID: 27578690
- Driver JA, Logroscino G, Gaziano JM, Kurth T. Incidence and remaining lifetime risk of Parkinson disease in advanced age. Neurology. 2009; 72:432–8. <u>https://doi.org/10.1212/01.wnl.0000341769.50075.</u> bb PMID: 19188574
- Kyrozis A, Ghika A, Stathopoulos P, Vassilopoulos D, Trichopoulos D, Trichopoulou A. Dietary and lifestyle variables in relation to incidence of Parkinson's disease in Greece. Eur J Epidemiol. 2013; 28:67– 77. https://doi.org/10.1007/s10654-012-9760-0 PMID: 23377703
- Driver JA, Smith A, Buring JE, Gaziano JM, Kurth T, Logroscino G. Prospective cohort study of type 2 diabetes and the risk of Parkinson's disease. Diabetes Care. 2008; 31:2003–5. <u>https://doi.org/10.2337/ dc08-0688 PMID: 18599528</u>
- Hsu CL, Voss MW, Best JR, Handy TC, Madden K, Bolandzadeh N, et al. Elevated body mass index and maintenance of cognitive function in late life: exploring underlying neural mechanisms. Front Aging Neurosci. 2015; 7:155. https://doi.org/10.3389/fnagi.2015.00155 PMID: 26347646
- Craft S, Watson GS. Insulin and neurodegenerative disease: shared and specific mechanisms. Lancet Neurol. 2004; 3:169–78. https://doi.org/10.1016/S1474-4422(04)00681-7 PMID: 14980532
- De Pablo-Fernández E, Breen DP, Bouloux PM, Barker RA, Foltynie T, Warner TT. Neuroendocrine abnormalities in Parkinson's disease. J Neurol Neurosurg Psychiatry. 2017; 88(2):176–85. <u>https://doi.org/10.1136/jnnp-2016-314601</u> PMID: 27799297
- Howells DW, Porritt MJ, Wong JYF, Batchelor PE, Kalnins R, Hughes AJ, et al. Reduced BDNF mRNA expression in the Parkinson's disease substantia nigra. Exp Neurol. 2000; 166:127–35. https://doi.org/ 10.1006/exnr.2000.7483 PMID: 11031089
- Nagahara AH, Tuszynski MH. Potential therapeutic uses of BDNF in neurological and psychiatric disorders. Nat Rev Drug Discov. 2011; 10:209–19. https://doi.org/10.1038/nrd3366 PMID: 21358740
- BDNF Study Group. A controlled trial of recombinant methionyl human BDNF in ALS: the BDNF Study Group (phase III). Neurology. 1999; 52:1427–33. PMID: 10227630

- Pickrell AM, Youle RJ. The roles of PINK1, parkin, and mitochondrial fidelity in Parkinson's disease. Neuron. 2015; 85:257–73. https://doi.org/10.1016/j.neuron.2014.12.007 PMID: 25611507
- Bové J, Martínez-Vicente M, Vila M. Fighting neurodegeneration with rapamycin: mechanistic insights. Nat Rev Neurosci. 2011; 12:437–52. https://doi.org/10.1038/nrn3068 PMID: 21772323
- Xu Q, Park Y, Huang X, Hollenbeck A, Blair A, Schatzkin A, et al. Physical activities and future risk of Parkinson disease. Neurology. 2010; 75:341–8. https://doi.org/10.1212/WNL.0b013e3181ea1597 PMID: 20660864
- Pontzer H, Durazo-Arvizu R, Dugas LR, PLange-Rhule J, Bovet P, Forrester TE, et al. Constrained total energy expenditure and metabolic adaptation to physical activity in adult humans. Curr Biol. 2016; 26:410–7. https://doi.org/10.1016/j.cub.2015.12.046 PMID: 26832439
- Richmond RC, Davey Smith G, Ness AR, den Hoed M, McMahon G, Timpson NJ. Assessing causality in the association between child adiposity and physical activity levels: a Mendelian randomization analysis. PLoS Med. 2014;11(3):e1001618. https://doi.org/10.1371/journal.pmed.1001618 PMID: 24642734
- Østergaard SD, Mukherjee S, Sharp SJ, Proitsi P, Lotta LA, Day F, et al. Associations between potentially modifiable risk factors and Alzheimer disease: a Mendelian randomization study. PLoS Med. 2015;12(6):e1001841. https://doi.org/10.1371/journal.pmed.1001841 PMID: 26079503
- Lawlor DA. Commentary: two-sample Mendelian randomization: opportunities and challenges. Int J Epidemiol. 2016; 43(3):908–15.
- 39. Tyrrell J, Jones SE, Beaumont R, Astley CM, Lovell R, Yaghootkar H, et al. Height, body mass index, and socioeconomic status: Mendelian randomisation study in UK Biobank. BMJ. 2016; 352:i582. https://doi.org/10.1136/bmj.i582 PMID: 26956984
- 40. Sudlow C, Gallacher J, Allen N, Beral V, Burton P, Danesh J, et al. UK Biobank: an open access resource for identifying the causes of a wide range of complex diseases of middle and old age. PLoS Med. 2015;12(3):e1001779. https://doi.org/10.1371/journal.pmed.1001779 PMID: 25826379
- Lauby-Secretan B, Scoccianti C, Loomis D, Grosse Y, Bianchini F, Straif K, et al. Body fatness and cancer—viewpoint of the IARC working group. N Engl J Med. 2016; 375:794–8. <u>https://doi.org/10.1056/ NEJMsr1606602</u> PMID: 27557308
- 42. Holmes MV, Dale CE, Zuccolo L, Silverwood RJ, Guo Y, Ye Z, et al. Association between alcohol and cardiovascular disease: Mendelian randomisation analysis based on individual participant data. BMJ. 2014; 349:g4164. https://doi.org/10.1136/bmj.g4164 PMID: 25011450